Author's response to reviews

Title: MS in South Asians in England: early disease onset and novel pattern of myelin autoimmunity

Authors:

Richard S Nicholas (r.nicholas@imperial.ac.uk)
Vassiliki Koastadima (vkostadima@yahoo.com)
Maya Hanspal (mah83@cam.ac.uk)
Benjamin R Wakerley (benwakerley@fastmail.fm)
Ruhena Sergeant (Ruhena.Sergeant@imperial.nhs.uk)
Saskia deCuypere (sdecuypere@ichr.uwa.edu.au)
Omar Malik (malikomar@hotmail.com)
Rosemary J Boyton (r.boyton@ic.ac.uk)
Daniel M Altmann (d.altmann@ic.ac.uk)

Version: 2 Date: 1 December 2014

Author's response to reviews:

Dear Nhell

We have revised the submissions as advised. There were no resources used here that were listed on SciCrunch.com

Best wishes

Danny Altmann